Author’s response to reviews

Title: Multi-parametric Muscle and Fat Correlation of Computed Tomography Parameters to Outcomes in a Total Hip Arthroplasty Population

Authors:

Michael Heffler (michael.heffler@utsouthwestern.edu)
Ryan Barlow (ryan.barlow@utsouthwestern.edu)
Yin Xi (yin.xi@utsouthwestern.edu)
Daichi Hayashi (daichi_hayashi@hotmail.com)
Hayden Box (hayden.box@phhs.org)
Michael Huo (michael.huo@utsouthwestern.edu)
Avneesh Chhabra (avneesh.chhabra@utsouthwestern.edu)

Version: 3 Date: 21 Dec 2017

Author’s response to reviews:

1. Email addresses have been added to the title page, and the corresponding author indicated.

2. Full names of the IRBs involved as well as study reference numbers have been added to the Declarations section. Additionally information clarifying consent procedures has been added.

3. Editorial positions of AC and DH have been added to the declarations section.

4. Information justifying contributions of DH to warrant authorship have been added. "DH provided guidance for improvement of the manuscript content by revising it critically and advising inclusion of new, intellectually important content. The manuscript was edited by
MAH, RB, AC, and DH. The manuscript was reviewed and approved for final publication by all authors.

5. The manuscript has been uploaded as requested with no tracked changes.